

Cáceres, 14th September of 2018

Dear Editor of World Journal of Hematology.

First, I would like to apologize for the delay in answering to the reviewers issues; I was out of office for holidays with limited access to internet.

I would like to thanks the reviewers their positive comments respect to my manuscript as well as to their suggestions in order to become a better quality manuscript.

As suggested by the reviewers I contacted one of my Native English collaborators, Dr. Steward O Sage, from the University of Cambridge (UK), in order to perform a English improving of my manuscript. I wish it would be now ready for being accepted and published in your journal.

Best Regards

Pedro C Redondo

Answers to reviewers.

Reviewer 1.

**“This manuscript deals with a timely and relevant issue, namely the biology of platelets with regard to miRNA regulatory functions. Presenting this short survey of the pertinent literature in order to raise awareness of the importance of the issues discussed therein is appropriate for an Editorial. Having said that, I must point out that the manuscript must be thoroughly revised as to English usage in order to make it useful to the Journal's readership. I strongly recommend that it be submitted to a native speaker of English.”**

- I apologies for the mistakes detected, the manuscript has been corrected by a native English speaker.

Reviewer 2.

**“The author summarizes the physiological and pathological role of miRNA regulating platelet function. It is well described and educational.”**

- I would like to thank the reviewer his/her nice comment on my manuscript, I have now addressed the English issue.

Reviewer 3.

**“The minireview y Pedro Redondo about the role of miRNA in megakaryocytes and platelet biology is a well written, up-dated and comprehensive article. References should have all the same style. Many of them do not have the title of the article, please unify all references. Please, correct grammatical errors.”**

- I thank the reviewer his/her comments, and I have now addressed all the issues raised against my manuscript.